Author’s response to reviews

Title: Addressing malnutrition among children in routine care: How is the Integrated Management of Childhood Illnesses strategy implemented at health centre level in Burundi?

Authors:

Manassé NIMPAGARITSE (mnimpagaritse@yahoo.fr)
Catherine Korachais (ckorachais@itg.be)
Georges Nsengiyumva (georges.nsengiyumva@gmail.com)
Jean Macq (jean.macq@uclouvain.be)
Bruno Meessen (bmeessen@itg.be)

Version: 3 Date: 22 Nov 2018

Author’s response to reviews:

Editor Comments:

Currently you have stated, the funding body covered the design; collection, analysis, and interpretation of data of the impact evaluation. Whilst we understand that the funding provided expenditure for the design etc, we are asking about the ROLE of the funder in the collection, analysis, and interpretation of data of the impact evaluation. This section is similar to the competing interest and is more about whether the funder was directly involved in the processes above. Are the views in the article the authors or the views of the funders?

Please note that, as a standard, we require that all commercial sources of funding are mentioned in the Competing Interests section. You may choose to duplicate the funding statement in the Competing Interests section.

Author: We apologize for that

Addressed (Line 392-398)